Author’s response to reviews

Title: The GP Tests of Competence assessment: which part best predicts fitness to practise decisions?

Authors:

Hirosha Keshani Jayaweera (h.jayaweera@ucl.ac.uk)

Henry Potts (henry.potts@ucl.ac.uk)

Karim Keshwani (karim.keshwani@ucl.ac.uk)

Chris Valerio (chris.valerio@ucl.ac.uk)

Magdalen Baker (rmhamtb@ucl.ac.uk)

Leila Mehdizadeh (l.mehdizadeh@ucl.ac.uk)

Alison Sturrock (alison.sturrock@ucl.ac.uk)

Version: 3 Date: 13 Dec 2017

Author’s response to reviews:

We have adjusted the manuscript as per the reviewer recommendations. These changes are listed below:

1) Author Emails
These have been added to the title page as requested.

2) Abstract Headings
The heading within the abstract has been changed from 'Background and Purpose' to 'Background'

3) List of Abbreviations
A list of abbreviations has been added before the declarations section.

4) Declarations Heading
Declarations heading has been added following the conclusion section.

5) Ethics Section
Ethics section has been modified as per requested and details regarding consent not being necessary has been removed.
6) Availability of Data and Materials
This section has been added

7) Conflict of Interest
The word 'that' has been removed as requested.

8) Clean Manuscript
A clean manuscript has been provided.